Author's response to reviews

Title: Giant uterine artery pseudoaneurysm after a missed miscarriage termination in a cesarean scar pregnancy

Authors:

Yun Mou (xuyuntin@163.com)
Yuezhen Xu (xuyuezhen001@sina.com)
Ying Hu (2216315256@qq.com)
Tianan Jiang (chenny69@126.com)

Version: 5
Date: 14 July 2014

Author's response to reviews: see over
Dear Editor:

Thank you very much for your consideration of our manuscript. We have made revisions according your comments and suggestions. The paper has also been edited carefully to conform to the journal style.

We hope the manuscript is now suitable for publication in *BMC women’s health*, and we look forward to hearing from you at your earliest convenience.

Yours sincerely,

Tianan Jiang

China

Response to Editorial Requirements:

1. Case Presentation

Please include information about the patient’s ethnic background in your case presentation. This should also be included in your abstract.

**Author’s response:** Thank you very much for your comment. We have added “Chinese” in Abstract on page 2 line 32 to describe the ethnic background of the patient. On page 3 line 70, we added “Chinese” too.

2. Additional Files

Thank you for including the tracked changes copy of your manuscript and your certificate of editing. Please now remove these so that they are will not be published alongside your manuscript.

**Author’s response:** Thank you very much for your comment. We have removed them in revised version.